ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Leitner, Sandra M.; Pöschl, Johannes; Stehrer, Robert

Working Paper Change Begets Change: Employment Effects of Technological and Non-Technological Innovations - a Comparison across Countries

wiiw Working Paper, No. 72

Provided in Cooperation with:

The Vienna Institute for International Economic Studies (wiiw) - Wiener Institut für Internationale Wirtschaftsvergleiche (wiiw)

Suggested Citation: Leitner, Sandra M.; Pöschl, Johannes; Stehrer, Robert (2011) : Change Begets Change: Employment Effects of Technological and Non-Technological Innovations - a Comparison across Countries, wiiw Working Paper, No. 72, The Vienna Institute for International Economic Studies (wiiw), Vienna

This Version is available at: https://hdl.handle.net/10419/203937

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Wiener Institut für Internationale Wirtschaftsvergleiche The Vienna Institute for International Economic Studies

Working Papers | 72 | January 2011

Sandra M. Leitner, Johannes Pöschl and Robert Stehrer

Change Begets Change: Employment Effects of Technological and Non Technological Innovations – a Comparison across Countries



wiiw Working Papers published since 2007

- No. 72 S. M. Leitner, J. Pöschl and R. Stehrer: Change begets change: Employment effects of technological and nontechnological innovations – A comparison across countries. January 2011
- No. 71 M. Holzner: Inequality, Growth and Public Spending in Central, East and Southeast Europe. October 2010
- No. 70 N. Foster, J. Pöschl and R. Stehrer: The Impact of Preferential Trade Agreements on the Margins of International Trade. September 2010
- No. 69 L. Podkaminer: Discrepancies between Purchasing Power Parities and Exchange Rates under the Law of One Price: A Puzzle (partly) Explained? September 2010
- No. 68 K. Hauzenberger and R. Stehrer: An Empirical Characterization of Redistribution Shocks and Output Dynamics. August 2010
- No. 67 R. Stöllinger, R. Stehrer and J. Pöschl: Austrian Exporters: A Firm-Level Analysis. July 2010
- No. 66 M. Holzner: Tourism and Economic Development: the Beach Disease? June 2010
- No. 65 A. Bhaduri: A Contribution to the Theory of Financial Fragility and Crisis. May 2010
- No. 64 L. Podkaminer: Why Are Goods Cheaper in Rich Countries? Beyond the Balassa-Samuelson Effect. April 2010
- No. 63 K. Laski, J. Osiatynski and J. Zieba: The Government Expenditure Multiplier and its Estimates for Poland in 2006-2009. March 2010
- No. 62 A. Bhaduri: The Implications of Financial Asset and Housing Markets on Profit- and Wage-led Growth: Some Results in Comparative Statics. February 2010
- No. 61 N. Foster and R. Stehrer: Preferential Trade Agreements and the Structure of International Trade. January 2010
- No. 60 J. Francois and B. Hoekman: Services Trade and Policy. December 2009
- No. 59 C. Lennon: Trade in Services: Cross-Border Trade vs. Commercial Presence. Evidence of Complementarity. November 2009
- No. 58 N. Foster and J. Pöschl: The Importance of Labour Mobility for Spillovers across Industries. October 2009
- No. 57 J. Crespo-Cuaresma, G. Doppelhofer and M. Feldkircher: The Determinants of Economic Growth in European Regions. September 2009
- No. 56 W. Koller and R. Stehrer: Trade Integration, Outsourcing and Employment in Austria: A Decomposition Approach. July 2009
- No. 55 U. Schneider and M. Wagner: Catching Growth Determinants with the Adaptive Lasso. June 2009
- No. 54 J. Crespo-Cuaresma, N. Foster and R. Stehrer: The Determinants of Regional Economic Growth by Quantile. May 2009
- No. 53 C. Lennon: Trade in Services and Trade in Goods: Differences and Complementarities. April 2009
- No. 52 J. F. Francois and C. R. Shiells: Dynamic Factor Price Equalization and International Convergence. March 2009
- No. 51 P. Esposito and R. Stehrer: Effects of High-Tech Capital, FDI and Outsourcing on Demand for Skills in West and East. March 2009
- No. 50 C. Fillat-Castejón, J. F. Francois and J. Wörz: Cross-Border Trade and FDI in Services. February 2009
- No. 49 L. Podkaminer: Real Convergence and Inflation: Long-Term Tendency vs. Short-Term Performance. December 2008
- No. 48 C. Bellak, M. Leibrecht and R. Stehrer: The Role of Public Policy in Closing Foreign Direct Investment Gaps: An Empirical Analysis. October 2008
- No. 47 N. Foster and R. Stehrer: Sectoral Productivity, Density and Agglomeration in the Wider Europe. September 2008
- No. 46 A. Iara: Skill Diffusion by Temporary Migration? Returns to Western European Work Experience in Central and East European Countries. July 2008
- No. 45 K. Laski: Do Increased Private Saving Rates Spur Economic Growth? September 2007
- No. 44 R. C. Feenstra: Globalization and Its Impact on Labour. July 2007
- No. 43 P. Esposito and R. Stehrer: The Sector Bias of Skill-biased Technical Change and the Rising Skill Premium in Transition Economies. May 2007
- No. 42 A. Bhaduri: On the Dynamics of Profit- and Wage-led Growth. March 2007

Sandra M. Leitner and Johannes Pöschl are economists at the Vienna Institute for International Economic Studies (wiiw). Robert Stehrer is wiiw Deputy Director of Research.

This publication was produced in the framework of MI-CRO-DYN (www.micro-dyn.eu), an international economic research project focusing on the competitiveness of firms, regions and industries in the knowledge-based economy. The project is funded by the EU Sixth Framework Programme (www.cordis.lu) with socio-economic sciences and humanities (http://ec.europa.eu/research/socialsciences/index_en.html).

This publication reflects only the authors' views, the European Community is not liable for any use that may be made of the information contained therein.





Sandra M. Leitner, Johannes Pöschl and Robert Stehrer

Change Begets Change: Employment Effects of Technological and Non Technological Innovations – a Comparison across Countries

Contents

Abs	tract		i
1.	Introc	duction	1
2.	Empi	rical evidence	3
3.	Mode	9	6
4.	Data		10
5.	Resu	lts	12
	5.1	Overall pool of countries	12
	5.2	By regions	15
6.	Sumr	mary and conclusion	17
7.	Refer	rences	19
8.	Appe	ndix	21
	8.1	Regressors	21
	8.2	Instruments	

List of Tables and Figures

Table 1	Overall sample	11
Table 2	Effects of product and process innovations on employment in Manufacturing and Services, by country-groups	13
Table 3	Effects of product and process innovations on employment in Manufacturing and Services, by country-groups (GMM)	22

Abstract

This paper attempts to shed new light on the effect innovation has on employment. Specifically, it identifies the net employment effects of technological product and process innovations as well as complementary non-technological organizational innovations which have so far mostly been bypassed in comparable analyses. The analysis applies the 4th Community Innovation Survey and determines and compares innovation-induced employment effects in both manufacturing and service sectors across three country-groups: i) a set of Central and Eastern European transition countries, ii) a group of Southern EU member states as well as iii) a pool of Core EU member countries. The results reveal interesting differences across types of innovation, sectors or country-groups analyzed. Particularly, in both manufacturing and service sectors of Central and Eastern European transition countries and Southern European countries employment expands in response to the introduction of product novelties or process innovations only. Non-technological organizational innovations, on the other hand, had a detrimental effect on employment in the manufacturing sector of Central and Eastern European countries only. In contrast, employment in both manufacturing and service sectors of Core European countries only reacts to the introduction of new products but remains unaffected by the implementation of process or organizational innovations.

Keywords: employment, technological and non-technological innovations, manufacturing and services, CIS 4

JEL classification: J2, O33

Change begets change: employment effects of technological and non-technological innovations – a comparison across countries

"In the 17th century nearly all Europe experienced revolts of the workpeople against the ribbon-loom, a machine for weaving ribbons and trimmings, called in Germany Bandmühle, Schnurmühle, and Mühlenstuhl. These machines were invented in Germany. Abbé Lancellotti, in a work that appeared in Venice in 1636, but which was written in 1579, says as follows:

'Anthony Müller of Danzig saw about 50 years ago in that town, a very ingenious machine, which weaves 4 to 6 pieces at once. But the Mayor being apprehensive that this invention might throw a large number of workmen on the streets, caused the inventor to be secretly strangled or drowned.'

In Leyden, this machine was not used till 1629; there the riots of the ribbon-weavers at length compelled the Town Council to prohibit it."

(Karl Marx in "Capital. *A Critique of Political Economy*", Volume I Book One: "The Process of Production of Capital", 1977, Chapter 15, p 283)

1. Introduction

Whether technological change tends to create or destroy employment has been an issue of scientific concern and policy debate for decades now. Particularly, ever since the Industrial Revolution the world has witnessed the creation of novel products and processes at an ever increasing pace. And while a positive long-term effect on aggregate employment seems apparent, the precise impact on employment at the firm level is less well known.

Theoretically, product and process innovations may both create and destroy jobs, rendering the net effect an a priori unclear outcome. Specifically, newly introduced products are likely to increase overall demand for a firm's products and, as a consequence, stimulate production and increase employment. This *labor compensation effect* is stronger the lower the degree of competition innovative firms face on their markets and the lower prevailing production synergies between old and new products. However, the demand for labor may fall should the product novelty replace a firm's old products. Furthermore, additional *labor displacement effects* may surface as, even in the absence of accompanying process innovations, product innovations tend to give rise to labor-saving productivity effects as a result of changes in production methods or input mixes. Process innovations, on the other hand, tend to reduce employment – at least in the short run - as, in response to associated productivity improvements, overall factor input requirements fall which allows process innovators to produce the same amount of output with fewer employees and lower unit costs. The exact scale of this *labor displacement effect* depends on the degree of substitutability of input factors and the labor-saving nature of the production technologies utilized. Moreover, successful process innovators may pass on cost reductions to consumers and reduce prices which, in turn, spurs demand and production and increases the demand for labor. This countervailing *labor compensation effect* is stronger the fiercer competition on the goods market as process innovators attempt to capture additional market shares and, at least temporarily, outpace their competitors by undercutting their prices.

Surprisingly however, empirical analyses have so far almost without exception bypassed potential employment effects of non-technological organizational innovations.¹ Innovative firms not only develop and introduce product and process innovations but also continuously adopt and reorganize organizational routines or practices and external relations so as to improve the utilization or exchange of information, knowledge or skills, the efficiency of work flows or the relations with other partners. Such organizational innovations which (1) are complementary to ICT investments, require almost no R&D resources for their development (Polder et al., 2010) but partly substantial investments in training activities (Antonelli et al., 2010), (2) tend to be implemented when real output prices or productivity is dwindling (Nickell et al., 2001), (3) are found to be complementary to product and process innovations (Cozzarin and Perzival, 2006, Polder et al., 2010, Schmidt and Rammer, 2007) and (4) tend to magnify the gains obtained through the introduction of technological product and process innovations (Faria and Lima, 2009) also affect labor productivity and employment. Specifically, some organizational innovations are labor-saving in that their strong labor-productivity enhancing effect decreases labor demand and employment. In that respect, such labor-saving organizational innovations are similar to process innovations. Other organizational innovations are capital-saving in nature and reduce the amount of material inputs required without cutting down on employment.

The ensuing analysis contributes to the ongoing discussion by identifying the net employment effects of technological product and process innovations as well as complementary non-technological organizational innovations in both manufacturing and service sectors of a diverse pool of European countries. Specifically, the analysis applies the 4th Community Innovation Survey for 14 different European countries, including six Central and Eastern European transition countries (the Czech Republic, Estonia, Hungary, Latvia, Slovenia and Slovakia which joined the EU in 2004), four Southern EU-member states (Spain, Greece, Italy and Portugal) and four Core EU-member countries (Denmark, France, Luxembourg

¹ Organizational innovation refer to new elements introduced into an organization's production or service operations - input materials, task specifications, equipment used to produce a product or render a service (Damanpour, 1991).

and Sweden). Methodologically, a modified version of Jaumandreu's (2003) productionfunction framework is used. Moreover, the analysis explicitly accounts for employment effects of organizational innovations. It thereby addresses the obvious gap in the literature and provides evidence for an almost neglected innovation-employment nexus.

Overall, the results demonstrate that innovation-induced employment responses differ partly substantially across sectors or country-groups considered. Particularly, both manufacturing and service sectors of Central and Eastern European transition countries and Southern European countries display very similar employment responses to technological innovations. Employment expands as the result of the introduction of product novelties or of process innovations only, but shrinks as a consequence of jointly introduced product and process innovations. However, non-technological organizational innovations appear to have destroyed employment in the manufacturing sector of Central and Eastern European countries only which may be traced back to the widespread implementation of labor-saving managerial or relational organizational innovations like outsourcing or subcontracting. In contrast, hardly any innovation-induced employment responses come to light in Core European countries. And while both manufacturing and service sector employment expand in the course of the introduction of a new product, practically no employment effects result from either process or organizational innovations.

The rest of the paper is organized as follows: Section 2 presents and discusses some previous research on the employment effects of product, process as well as non-technological organizational innovations while section 3 outlines the underlying theoretical model and addresses some emerging statistical issues. A discussion of the data used in the empirical analysis is provided in section 4. Section 5 presents the main empirical results for both the overall sample as well as separately for the three country-groups considered while section 6 concludes.

2. Empirical evidence

The question as to whether technological change affects employment is an old one (Say, 1803) and has aroused interest of economists and politicians alike. Since then a rich strand of literature has emerged which has attempted to identify the employment effects of product and process innovations or, to a lesser extent, the employment effects of complementary non-technological organizational innovations.

All in all, conclusive evidence is found that *product innovations* are associated with employment growth at the firm level (Van Reenen, 1997, Smolny, 1998, Rottmann and Ruschinski, 1998, Lachenmaier and Rottmann, 2006, Becker and Ecker, 2007, Garcia et al., 2002, Hall et al., 2007, or Benavente and Lauterbach, 2008). A less optimistic picture is drawn by Klette and Forre (1998) in their study on a panel of Norwegian manufacturing

plants which is unable to discover any straightforward positive relationship between innovation and net job creation.

More recently, Community Innovation Surveys (CIS) became subject of similar analyses. In contrast to firm panels, these Surveys are representative of the overall economy and harmonized with renders them comparable across countries. Moreover, unlike most of the above-cited studies which almost exclusively focus on the manufacturing sector, Community Innovation Surveys are not confined to the manufacturing sector and therefore account for the increasing importance of business R&D and innovation reported for the service sector (Sirilli and Evangelista, 1998; Evangelista, 2000). Empirical results underscore the importance of product innovations in stimulating employment. Whether the expected employment effect of product innovations differs by employees' skill-level is addressed by Falk (1999) who focuses on West German innovators as covered by the 2nd Community Innovation Survey. He concludes that firstly, university graduates experience the strongest employment growth in response to the implementation of product innovations and that secondly, irrespective of employees' skill-level, the joint implementation of new products and processes results in higher overall employment growth as compared to the implementation of either product or process innovations only.

First evidence on the effect innovation has on employment in the service sector is provided by Evangelista and Savona (2003) and Jaumandreu (2003). Specifically, Evangelista and Savona (2003) use information collected by the Italian Innovation Survey and show that firms which introduce service innovations are more likely to undergo positive employment changes. They also demonstrate that the effect differs by employees' skill-levels as product innovations are more likely to stimulate employment of highly skilled employees while no significant employment effect emerges for low-skilled employees. Moreover, Jaumandreu (2003) develops a production-function framework which allows him to disentangle the labor displacement and compensation effects of innovations. His analysis which is based on the 3rd Spanish Community Innovation Survey highlights that product innovations generate positive net employment effects in both the manufacturing and service sectors.

Similar analyses were conducted by Harrison et al. (2005) and Peters (2004) which also rest on the methodology developed by Jaumandreu (2003). Specifically, Harrison et al. (2005) take a multi-country approach to identify and compare innovation-related employment effects of four of the largest European economies: France, Germany, Spain and the UK. They stress that while, in general, product innovations are associated with positive net employment effects in both the manufacturing and service sector, interesting cross-country and cross-sectoral differences emerge: Firstly, product innovations appear to have the strongest role in stimulating employment growth in Germany and secondly, employment effects are generally not stronger in the manufacturing sector. Instead, the service sectors

of France and the UK experience higher innovation-related employment growth while more pronounced effects become apparent in the manufacturing sectors of Germany and Spain.

That employment effects may differ by the degree of novelty of product innovations is taken up by Peters (2004). Specifically, she seeks to identify in how far the introduction of market novelties, as compared to the introduction of firm novelties only, is related to more pronounced employment effects. The analysis uses the 3rd German Community Innovation Survey and shows that the degree of novelty of the product innovation hardly matters since in both the manufacturing and service sectors alike employment effects are positive but not significantly different from each other.

And that technological leadership matters little is suggested by the findings of Meriküll (2008). She applies the 3rd and 4th Community Innovation Surveys for Estonia and high-lights that even for a technologically lagging economy like Estonia product innovations are associated with positive overall employment effects.

In contrast, however, results are mixed and less clear-cut for employment effects of *process innovations*. Specifically, positive effects are found by Smolny (1998), Lachenmaier and Rottmann (2006) and Becker and Egger (2007) for West German firms or Garcia et al. (2002) for a set of Spanish firms. A negative employment effect is identified by Ross and Zimmermann (1993) for German manufacturing firms. Finally, Van Reenen (1997) for UK manufacturing firms, Rottmann and Ruschinski (1998) for West German firms, Hall et al. (2007) for a panel of Italian firms and Benavente and Lauterbach for Chilean firms find no significant effect of process innovations on firm-level employment at all.

Moreover, CIS-based analyses reach similar mixed conclusions. That process innovations lead to the loss of employment in German manufacturing but remains insignificant in German services is highlighted by Peters (2004). Furthermore, she also takes into account that process innovations are introduced for various reasons, potentially entailing very different employment effects. Specifically, the employment effects of two different types of process innovations are analyzed: process innovation intended to reduce production costs – so called rationalization innovations - on the one hand, and process innovations intended to improve product quality, increase production capacity or to reduce personnel costs, on the other. The results highlight that only manufacturing firms which introduced process innovations for rationalization purposes significantly reduced employment while no such effect is apparent for service firms.

Finally, no significant effects are found by Harrison et al. (2008) for France, Germany and the UK and Meriküll (2008) for Estonia while slightly positive employment effects of process innovations emerge in Spanish manufacturing (Harrison et al., 2008).

However, in contrast to comprehensive empirical evidence on the employment effects of both product and process innovations, only very few studies have examined the likely effect the implementation of non-technological *organizational innovations* has on firm-level employment. This can partly be traced back to the fact that no clear or uniform definition exists of what organizational innovations really are or how to best quantify their effects on output, productivity or employment.

Greenan (1995) provides an early contribution to the discussion and stresses that laborsaving organizational innovations tend to have positive employment effects only if they are introduced together with technological process innovations.

Recently, Evangelista and Vezzani (2010) used the fourth Community Innovation Survey for a pool of countries including the Czech Republic, Spain, France, Italy, Portugal and Slovenia to demonstrate that the introduction of new or improved organizational practices intended to bring about efficiency and quality improvements stimulates employment. Hence, activities affected or modified by organizational innovations tend to become more labor intensive. Moreover, the effect is stronger in the manufacturing sector. However, if introduced together with productivity-enhancing process innovations, organizational innovations appear to have a labor-saving effect in the manufacturing sector only.

The service sector is center-stage in the study conducted by Falk (2001) who stresses that German service firms which adopted new organizational practices like total quality management (TQM) systems, certified ISO 9000, lean administration, flatter hierarchies, delegation of authority and ICT-enabled organizational changes grew three percent points faster than non-adopters. Moreover, no discernible differences can be identified across the different new organizational practices analyzed.

3. Model

Following Jaumandreu (2003), a firm can produce two different types of products: an old or only marginally modified or improved product – labeled *'old product'* - as well as a new or significantly improved product – labeled *'new product'*. The output of the old product is denoted by Y_{1t} while the output of the new product is denoted by Y_{2t} .

Firms are observed at two points in time, at the beginning of the period (t = 1) as well as at the end of the period (t = 2). At the beginning of the period, all products are old products and in the course of the period the firm decides whether to introduce a new product. Each type of product is produced with an identical constant-returns-to-scale production technology

$$Y_{it} = \theta_{it} F(K_{it}, L_{it}, M_{it}) \quad \text{with } i = 1, 2 \text{ and } t = 1, 2, \qquad (1)$$

where θ_{it} represents an efficiency parameter which captures productivity improvements of all inputs used, K_{it} refers to capital, L_{it} stands for labor and M_{it} denotes materials. Production efficiencies θ_{it} of either type of product are variable over time and change should the firm decide to invest in and implement productivity-enhancing process or organizational innovations. Generally, production efficiencies of old and new products can differ in that new products may be produced more efficiently than old products such that $\theta_{2t} > \theta_{1t}$, or vice versa such that $\theta_{2t} < \theta_{1t}$.

These production functions for old and new products correspond to the following cost functions at time t:

$$C(w_{1t}, w_{2t}, Y_{1t}, Y_{2t}, \theta_{1t}, \theta_{2t}) = C(w_{1t}) \frac{Y_{1t}}{\theta_{1t}} + C(w_{2t}) \frac{Y_{2t}}{\theta_{2t}} + F$$
(2)

where $c(w_{it})$ refers to marginal costs of either old or new products while F denotes fixed costs. Applying Shepard's Lemma which states that the partial derivative of expenditurefunctions with respect to the price gives the Hicksian demand function, labor demand at time t can be derived as:

$$L_{it} = c_L(w_{it}) \frac{Y_{it}}{\theta_{it}} , \qquad (3)$$

where $c_L(w_{it})$ is the derivative of marginal costs with respect to wages. Hence, labor demand for the production of either old or new products is higher the higher the product's output Y_{it} and the lower the product's production efficiency θ_{it} .

Employment growth over the entire period is defined as follows

$$\frac{\Delta L}{L} = \frac{\Delta L_1 + \Delta L_2}{L_1} = \frac{L_{12} - L_{11}}{L_{11}} + \frac{L_{22} - L_{21}}{L_{11}} = \frac{L_{12} - L_{11}}{L_{11}} + \frac{L_{22}}{L_{11}}, \tag{4}$$

as the sum of employment growth due to the production of the old product and of employment growth due to the production of the new product. Since production of the new product only commences in the course of the period but is still zero at the beginning of the period, labor demand at t = 1 is $L_{21} = 0$.

As outlined above, production efficiencies of old products θ_1 are not constant but are subject to change. Specifically, overall efficiency changes of old products $\Delta \theta_1$ either result from efficiency changes due to process innovations $\Delta \delta_1$ or from efficiency changes due to non-technological organizational innovations $\Delta \omega$, so that $\Delta \theta_1 = \Delta \delta_1 + \Delta \omega$. Together with constant wages w, and $L_{11} = c_L(w) \frac{Y_1}{\theta_1}$ denoting employment for the production of the old production of the old product at the end of the period and $L_{22} = c_L(w) \frac{Y_2}{\theta_2}$ denoting employment for the production of the new product at the end of the period, equation (4) then becomes:

$$\frac{\Delta L}{L} = \left(\frac{\mathbf{Y}_1 + \Delta \mathbf{Y}_1}{\theta_1 + \Delta \theta_1}\right) \left(\frac{\theta_1}{\mathbf{Y}_1}\right) - 1 + \frac{\mathbf{Y}_2}{\mathbf{Y}_1} \frac{\theta_1}{\theta_2} \cong -\frac{\Delta \delta_1}{\theta_1} - \frac{\Delta \omega}{\theta_1} + \frac{\Delta \mathbf{Y}_1}{\frac{\mathbf{Y}_1}{2}} + \frac{\mathbf{Y}_2}{\frac{\mathbf{Y}_1}{3}} + \left(\frac{\mathbf{Y}_2}{\mathbf{Y}_1}\right) \left(\frac{\theta_1}{\theta_2}\right).$$
(5)

Hence, with constant input prices, employment changes are the result of four forces: 1) the change in the production efficiency in the production of the *old product* stemming from process innovations, 2) the change in the production efficiency in the production of the *old product* arising from organizational innovations, 3) the change in the output of the *old product* and 4) the relative output and efficiency of the *new product* (relative to the old product).

Improvements in the production efficiency of old products result from both newly implemented process innovations as well as organizational change in terms of new or improved routines and practices. However, efficiency improvements in the production of old products not only accrue to firms which actually implement process and organizational innovations in their production processes but also arise to firms which abstain from adopting such innovations at all. But any such efficiency improvements are expected to be higher for process and organizational innovators while efficiency gains of non-process/organizational innovators are expected to be more moderate, predominantly resulting from exogenous technological change, learning-by-doing or spillover effects.

Moreover, the employment effect of the production of new products depends on the relative efficiencies of producing old and new products. And if the production efficiency of the new product θ_2 is higher than the production efficiency of the old product θ_1 , so that $(\theta_1/\theta_2) < 1$, then a labor-displacement effect emerges and employment does not grow one-for-one with output growth accounted for by the new product.

(6)

Equation (5) then translates into the following econometric specification:

$$I = \alpha_0 + \alpha_1 d_p + \alpha_2 d_o + y_1 + \beta y_2 + u$$

where:

- / refers to the employment growth rate,
- α_1 denotes the average efficiency growth of process innovators,
- d_{p} is a dummy variable for process innovations,
- α_2 denotes the average efficiency growth of organizational innovators,
- *d*_o is a dummy variable for organizational innovations,
- y_1 refers to real output growth due to old products,

 β captures the relative efficiency of the production of old and new products,

 y_2 refers to real output growth due to new products and

u is an error term with $E(u | d_p, d_o, y_1, y_2) = 0$.

However, real output growth is not observable in the data but instead needs to be approximated with nominal sales growth. Additionally, observed nominal sales growth is split up into nominal sales growth due to the old product (g_1) and into nominal sales growth due to the new product (g_2). Furthermore, with π_1 denoting price variations of the old product, nominal sales growth of the old product is approximately $g_1 = y_1 + \pi_1$. Moreover, denoting

 $\pi_2 = \frac{p_2 - p_1}{p_1}$ as the proportional difference of prices of new products with respect to the

old prices, sales growth of new products becomes approximately $g_2 = y_2 + \pi_2 y_2$. Substituting g_1 and g_2 for y_1 and y_2 in equation (6) gives

$$I - \boldsymbol{g}_1 = \alpha_0 + \alpha_1 \boldsymbol{d}_p + \alpha_2 \boldsymbol{d}_o + \beta \boldsymbol{g}_2 + \upsilon \,. \tag{7}$$

- $I-g_1$ refers to total employment growth minus growth of nominal sales of unchanged products,
- β captures the relative efficiency of the production of *old* and *new* products,
- g_2 refers to sales growth of *new* products and
- v is an error term with $E(\pi_2 | y_2) = 0$.

All other variables (α_1 , d_p , α_2 and d_o) are as specified in equation (6).²

Equation (7) then provides the net employment effect of product, process and organizational innovations.

However, the estimation of equation (7) is plagued by two issues. Firstly, an endogeneity problem is present since sales growth of new products g_2 is correlated with the error term $v_{.}$ To address this problem, an instrumental variable (IV) approach is applied which will guarantee that estimators are unbiased. Potential instruments for the endogenous variable must be related to sales growth of new products but unrelated to the error term. Several candidates were taken into account and based on Sargan-Hansen overidentification tests the following sets of instruments were used: *new to market* (as the market share of goods and services introduced during 2002 and 2004 that were new to the firm's market), *R&D intensity* (as the ratio of total (intramural and extramural) R&D expenditures to firm turnover in 2004.), *patent* (a dummy which takes the value 1 if the firm applied for a patent between 2002 and 2004) and *science* (a dummy which is 1 if institutional sources like universities or other higher education institutions or government or public research institutes were of 'high'

² For information on the variables' definitions please refer to the Appendix.

or 'medium' importance as sources of information for a firm's innovation activities) for analyses of the manufacturing sector. Moreover, *new to market, patent* and *continuous innovator* (a dummy which is 1 if the firm reported continuous R&D activities during 2002 and 2004) were applied as instruments in the analyses of the service sector.

Secondly, an identification problem arises since firm-level price changes are unobservable in the dataset and therefore incorporated in the error term v. To remedy this identification problem and to control for firm-level price changes, country-specific disaggregated 2-digit industry-level price indices $\tilde{\pi}_1$ are included in the analysis. Hence, the dependent variable then becomes $I - (g_1 - \tilde{\pi}_1)$, which captures total employment growth minus real sales growth of unchanged products, instead of $I - g_1$ (from equation (7)) which refers to total employment growth minus growth of nominal sales of unchanged products. Provided that price changes at the firm level do not deviate much from sectoral averages, this then allows for the identification of the average gross real productivity effect. The analysis pursued in section 5 will focus on total employment growth minus real sales growth of unchanged products ($I - (g_1 - \tilde{\pi}_1)$) only.

4. Data

The analysis uses the fourth wave of the Community Innovation Survey (CIS-4), covering the three-year period from 2002 to 2004. Generally, the Community Innovation Survey is a harmonized and periodic survey designed to monitor firms' innovative activities, progress or obstacles over a pre-specified period and covers product and process innovations.

Moreover, the CIS-4 also collects more detailed information on non-technological innovations like organizational and marketing innovations. Specifically, *organizational innovations* refer to (i) new or significantly improved knowledge management systems aimed at better using or exchanging information, knowledge and skills, to (ii) major changes to the organization of work such as changes in management structures and to (iii) new or significant changes in the firm's external relations through alliances, partnerships, outsourcing or subcontracting. *Marketing innovations*, on the other hand, capture either significant changes to the design or packaging of goods and/or services, or new or significant changes in sales or distribution methods. Additionally, for organizational innovations, also their effects are determined in terms of reducing the response time to customer and supplier needs, improving the quality of goods or services, reducing costs per unit of output and improving employee satisfaction and/or reducing employee turnover.

The Fourth Community Innovation Survey was carried out in 25 EU-Member States, some EU-candidate countries as well as Iceland and Norway. It was made available for scientific research at the EUROSTAT Safe Center in Luxembourg by 18 participating countries.

In order to avoid that results are distorted by outliers (like new entrants) and to guarantee comparability across country micro-data the analysis is restricted to incumbent firms only which were established before 2002 and excludes all firms with less than 15 employees. Furthermore, it focuses on the manufacturing sector (NACE 15-37) and the service sector (NACE 51-74) but excludes the mining and quarrying sector (NACE 10-14) which is comparatively small and of little significance in terms of the degree of technological innovative-ness and change.

As for the growth rate of prices, corresponding to the industry classifications and industrygroups used, two-digit industry-level price indices were applied. These were taken from the EU KLEMS database which provides comparable price indices across countries. However, no price indices were available for Bulgaria, Iceland, Norway and Romania which were therefore left out from the analysis. The ensuing analysis uses these anonymized microlevel data sets for 14 countries.

	Overa	ll sample			
	Non- innovators	Innovators	Total	Non- innovators (in %)	Innovators (in %)
CENTRAL AND EASTERN EUROP	PE (CEEC)				
Czech Republic	2,844	2,178	5,022	56.6	43.4
Estonia	602	722	1,324	45.5	54.5
Hungary	2,054	813	2,867	71.6	28.4
Latvia	1,515	265	1,780	85.1	14.9
Slovenia	934	537	1,471	63.5	36.5
Slovakia	1,290	669	1,959	6.6	34.2
Total	9,239	5,184	14,423	64.1	35.9
SOUTHERN EUROPE (SOUTH)					
Spain	7,491	6,951	14,442	51.9	48.1
Greece	61	289	350	17.4	82.6
Italy	9,823	5,053	14,876	66.0	34.0
Portugal	1,875	1,754	3,629	51.7	48.3
Total	19,250	14,047	33,297	57.8	42.2
CORE EUROPE (CORE)					
Denmark	0	690	690	0.0	100.0
France	10,086	7,731	17,817	56.6	43.4
Luxembourg	179	262	441	40.6	59.4
Sweden	933	1,348	2,281	40.9	59.1
Total	11,198	10,031	21,229	52.7	47.3
Overall sample (ALL)	39,687	29,262	68,949	57.6	42.4

Note: an innovator is a firm that either successfully introduced a new or improved product or process, had still ongoing innovative activities or had to abandon its innovative activities during the respective survey period (2002-2004).

Source: EUROSTAT

Table 1

All in all, the analysis covers about 69,000 firms operating in 14 different European countries, including six Central and Eastern European transition countries (the Czech Republic, Estonia, Hungary, Latvia, Slovenia and Slovakia which joined the EU in 2004), four Southern EU-member states (Spain, Greece, Italy and Portugal) and four Core European countries (Denmark, France, Luxembourg and Sweden). More specifically, in terms of regional distribution, some 14,400 firms (or 21 percent of the overall sample analyzed) were located in the eight Central and Eastern European transition countries, another 33,300 firms (or 48 percent of the overall sample) were located in the four Southern EU-member states, while 21,200 firms (or 31 percent of the overall sample) were based in the six Core European countries (Table 1).

Approximately 42 percent of all firms analyzed were innovators in the sense that they have either successfully introduced a new or improved product or process, had still ongoing innovative activities or had to abandon their innovative activities during the 2002-2004 survey period. The prevalence of innovators differs partly substantially across regions and countries analyzed: only about 36 percent of firms located in the New Member States of Central and Eastern Europe conducted any innovative activities while 42 percent and 47 percent of all firms operating in Southern Europe and Core Europe, respectively, pursued any innovative activities.

5. Results

Empirical results for the employment effects of different types of innovations are presented in Table 2 below for both the manufacturing and service sectors for the overall pool of countries as well as for three more homogeneous country-groups, applying an Instrumental Variable (IV) approach. All estimations were also repeated applying the General Method of Moments (GMM) technique which guarantees that, if heteroskedasticity is present, efficient estimates can be obtained. Results, which are qualitatively comparable to the ones based on the IV approach, are presented in Table 3 in the Appendix.

5.1 Overall pool of countries

Generally, the analysis of the overall pool of countries provides robust and consistent evidence that the successful introduction of a *new product* on a firm's market has positive employment effects, in both manufacturing and services sectors. Hence, the additional demand created by the new product leads to the expansion of the firm's production (and market) and, consequently, to higher employment. Moreover, all coefficients are smaller than one, suggesting that new products are produced more efficiently than old ones. And this productivity-induced labor-saving effect results in employment to grow less than onefor-one with growth of sales accounted for by new products. Interestingly though, any employment effects appear to be slightly higher – though not significantly - for the service sector which indicates that productivity improvements in the production of new products are lower in the service sector than in the manufacturing sector.

Table 2

Effects of product and process innovations on employment in Manufacturing and Services, by country-groups

	Manufacturing				Services			
Country group	ALL	CEEC	CORE	SOUTH	ALL	CEEC	CORE	SOUTH
Method	IV	IV	IV	IV	IV	IV	IV	IV
Dependent variable	$I-(g_1-\widetilde{\pi}_1)$				$I-(g_1-\widetilde{\pi}_1)$			
Sales growth - new products	0.621	0.624	0.455	0.661	0.691	0.911	0.888	0.418
	(7.60)***	(5.25)***	(2.01)**	(5.53)***	(8.52)***	(4.95)***	(5.82)***	(3.75)***
Process innovations only	2.397	3.164	-0.525	3.832	1.668	3.442	-0.677	3.382
	(3.76)***	(1.90)*	(0.52)	(4.57)***	(1.73)*	(1.41)	(0.48)	(2.17)**
Process and product innovations	-5.193	-7.453	-1.272	-6.565	-5.896	-9.302	-1.487	-10.041
	(8.32)***	(5.12)***	(1.39)	(7.36)***	(5.88)***	(3.89)***	(1.01)	(5.93)***
Organizational innovations	-1.197	-1.876	-0.904	-0.850	0.502	0.549	-0.475	1.874
	(2.62)***	(1.79)*	(1.25)	(1.32)	(0.71)	(0.36)	(0.45)	(1.57)
No. of observations	25,809	7,502	7,006	11,301	17,083	3,798	7,336	5,949
Sargan test	6.172	14.442	3.572	5.709	0.157	1.623	0.0252	4.663
degrees of freedom	(3)	(3)	(3)	(3)	(2)	(2)	(2)	(2)
p-value	0.1035	0.0024	0.3115	0.1267	0.9246	0.4442	0.0252	0.0972

Notes: All specifications for the manufacturing sector use new to the market, R&D intensity, patent, science as instruments while all specifications for the service sector use new to the market, patent and continuous innovator as instruments. Country and industry dummies are included in all specifications.

The pooled sample (ALL) consists of the Czech Republic, Denmark, Estonia, France, Greece, Hungary, Italy, Latvia, Luxembourg, Portugal, Slovenia, Slovakia, Spain and Sweden.

Central and Eastern European transition countries (CEEC) comprise the Czech Republic, Estonia, Hungary, Latvia, Slovenia and Slovakia.

The group of Core European countries (CORE) comprises Denmark, France, Luxembourg and Sweden.

Southern European countries (SOUTH) comprise Greece, Italy, Portugal and Spain.

Country and industry dummies are included in all regressions.

Absolute value of t statistics in parentheses

* significant at 10%; ** significant at 5%; *** significant at 1%

Moreover, the analysis sheds light on the employment effects associated with the introduction of *process innovations*, either in terms of (i) new or significantly improved methods of manufacturing or producing goods and services, of (ii) new or significantly improved logistics, delivery or distribution methods for inputs, goods or services or in terms of (iii) new or significantly improved supporting activities for processes. These observable process innovations may be associated with both old as well as new products. However, the model outlined above explicitly highlights the role of old products only to have relevant employment effects. Therefore, the overall group of process innovators was split up into one group comprising process innovators only and one group comprising product and process innovators. Hence, employment effects of process innovations are determined separately as a response to the implementation of *process innovations (of old products) only* on the one hand and as a response to the implementation of *both product and process innovations (of old and new products)*, on the other.

The results highlight that, on average, firms in both manufacturing and service sectors which introduced *process innovations only* grew and significantly expanded their labor forces, while firms which introduced both *process and product innovations jointly* drastically reduced employment. The strong employment compensation effect observable for process innovators only is indicative of a strong cost-reduction effect of process innovations that was passed on to lower output prices. This, in turn, stimulated the demand for old products and, consequently, employment in order to satisfy the resulting increase in demand. Generally, positive employment effects of *process innovations only* appear to be stronger in the manufacturing sector which might reflect that fiercer competition on markets for manufacturing commodities induced manufacturing firms to pass a higher fraction of cost reductions on to output prices.

Additionally, theoretical analyses and empirical evidence emphasize that process and product innovations tend to be complementary and therefore implemented jointly in order to exploit emerging synergy effects, minimize associated adjustment costs and maximize/optimize profits (Damanpour and Gopalakrishnan, 2001; Mantovani, 2006). The findings highlight that *jointly implemented product and process innovations* entail sizeable displacement effects and lead to substantial employment reductions. Observable displacement effects are slightly stronger in the service sector which, in contrast to the manufacturing sector, suffered higher employment losses due to the joint introduction of both product and process innovations.

Finally, the analysis also accounts for potential employment effects associated with *organizational innovations* in terms of either (i) new or significantly improved knowledge management systems, (ii) major changes to the organization of work or (iii) new or significant changes in the relations with other firms or public institutions. Apparently, in the manufacturing sector, organizational innovations entail significant employment displacement effects as firms tend to reduce employment in the course of the implementation of strategic reorganization measures. In contrast, no significant employment effects emerge in the service sector which implies that service firms implement organizational innovations without major lay-offs.

However, these findings refer to a very heterogeneous pool of economies, including countries with very diverse levels of economic or technological development. Hence, dynamics and specificities of individual countries or more homogeneous country-groups may remain obscured. Therefore, to unveil any hidden effects or dynamics, all sector specific analyses were also conducted for the following three more homogeneous groups of countries: 1) technologically and economically lagging Central and Eastern European transition countries, (2) economically lagging Southern EU-member countries and, (3) economically more advanced Core European countries. The results are discussed in subsection 5.2.

5.2 By regions

A cross-regional comparison of innovation-induced employment effects consistently demonstrates that successfully introduced product novelties entail positive employment effects, irrespective of sector considered. Except for the South, these effects tend to be stronger in the service sector which therefore benefited more from product innovations by generating more employment opportunities. Moreover, the elasticity of employment with respect to sales growth of new products is close to one in the service sectors of Central and Eastern European transition countries and of Core European countries. This indicates that old and new products were produced with similar efficiency levels so that employment in the service sectors of all Central and Eastern European transition countries and Core European countries analyzed grew by almost one-to-one with sales growth of new products. In contrast, more moderate employment expansions are associated with sales growth of new products were produced considerably more efficiently than old ones which then translated into employment growth in the manufacturing sector to substantially lag behind sales growth of new products.

Additionally, interesting cross-country and sectoral differences are revealed once employment effects of process innovations are analyzed. Specifically, process innovators located in Southern European countries which successfully introduced process innovations only considerably expanded employment in both the manufacturing and service sectors. This effect appears to be slightly stronger in the manufacturing sector which suggests that possibly due to fiercer competition in the market for manufacturing goods a stronger passthrough mechanism of productivity improvements to cost-reductions and to reductions in output prices prevailed. This in turn rendered products more attractive and strongly stimulated demand and employment. Moreover, sector-specific employment responses to the introduction of process innovations only are apparent in Central and Eastern European transition countries. Specifically, while successful process-only innovators operating in the manufacturing sector considerably expanded their workforces, no significant employment effects were observable in the service sector. However, the results for Central and Eastern European transition countries and Southern European countries stand in stark contrast to findings for Core European countries where, irrespective of sector considered, employment remained unresponsive to the introduction of process innovations only. Hence, unlike in the other two groups considered, process innovations only were unable to generate any additional employment opportunities in Core European countries. This, in turn, indicates that

any labor compensation effects emanating from higher labor demand in response to lower output prices is exactly compensated by any labor displacement effects resulting from a change in the input-factor requirements associated with newly implemented processes.

In contrast, however, firms which introduced *both process and product innovations* jointly drastically reduced their workforces. Hence, firms which introduce both types of innovations jointly were able to save labor. This labor-saving or labor displacement effect tends to be stronger in the service sector which suffered more pronounced innovation-induced employment losses. Interestingly, a comparison highlights that these negative employment effects emerge for Central and Eastern European transition countries and Southern European countries only while, in contrast, employment in Core European countries is unaffected by the simultaneous introduction of process and product innovations. Hence, Core European countries did not experience any destruction of employment due to jointly introduced process and product innovations.

Finally, the analysis also sheds light on potential employment effects associated with the introduction of organizational innovations. As outlined above, organizational innovations are either intended to (i) improve prevailing internal knowledge management systems, (ii) to modify the firm's organization of work or (iii) to change external relations with other firms or institutions. The findings demonstrate that except for the manufacturing sectors of Central and Eastern European transition countries which suffered non-negligible employment losses, no employment effects emerge in conjunction with organizational restructuring measures of firms. This is good news for employment in either manufacturing or service sectors of South European or Core European countries which remained unaffected by the implementation of organizational innovations. These marked differences in employment responses across country-groups and sectors can be traced back to the very nature of the different organizational innovations covered by the general indicator 'organizational innovations' and the labor and capital saving effects they each entail. Specifically, (managerial) organizational innovations like (i) improved knowledge management systems intended to ease the use and exchange of information, knowledge and skills within a firm tend to increase labor productivity and, consequently, save labor which leads to major personnel cut-backs. Additionally, (relational) organizational innovations in terms of (iii) changes in external relations with other firms or organizations through, for example, outsourcing or subcontracting tend to destroy jobs and reduce employment. In contrast, organizational innovations in terms of (ii) a reorganization of work are likely to enhance capital productivity only, leaving labor productivity and labor demand unaffected. Hence, above results may indicate that manufacturing firms located in Central and Eastern European countries predominantly implemented labor-saving managerial and relational organizational innovations while capital-saving innovations which modify the organization of work dominated in firms located in Southern European countries and Core European countries.

6. Summary and conclusion

Theoretically, technological change has all the potentials to create, but more worryingly, to also destroy employment. Empirical evidence seems to suggest that, in the long run, aggregate employment benefits from technological change. However, short-run employment responses at the firm-level are still less well known, a shortcoming this study seeks to remedy. Specifically, this analysis seeks to indentify the employment effects of different types of innovation in the manufacturing and service sectors of three specific country-groups within the European Union. It uses the 4th Community Innovation Survey covering the period 2002 to 2004 for a set of 14 countries including six economically and technologically lagging Central and Eastern European transition countries which joined the EU in 2004 (the Czech Republic, Estonia, Hungary, Latvia, Slovenia and Slovakia), four economically lagging Southern EU-member states (Spain, Greece, Italy and Portugal) and four economically and technologically more advanced Core European countries (Denmark, France, Luxembourg and Sweden).

Generally, empirical findings reveal interesting and diverse innovation-induced employment responses that differ partly greatly across sectors or country-groups considered. The results consistently demonstrate that, irrespective of country-group or sector considered, the successful introduction of *product novelties* stimulates employment. Hence, firms expand their workforces as demand for and production of their new product increases. Moreover, except for the Southern EU-member states, this effect tends to be stronger in the service sector which is indicative of less pronounced efficiency improvements in the production of new service products.

In contrast, however, employment effects are mixed and less clear-cut in response to either newly implemented process innovations or non-technological organizational innovations. Specifically, sizeable positive employment effects emerge in both manufacturing and service sectors of Southern European countries in conjunction with the introduction of *process innovations only*. Similar positive effects are apparent in the manufacturing sector of Central and Eastern European transition countries while no such effect exists for these countries' service sectors. All in all, these findings suggest that a strong pass-through mechanism of productivity improvements to cost-reductions and to reductions in output prices prevailed in these two country-groups. This, in turn, rendered new products more affordable and attractive, strongly stimulated demand, production and, consequently, employment. In contrast, however, process innovations only were unable to stimulate employment in either sector in Core European countries.

However, firms which introduced *both process and product innovations* jointly were able to save labor and to drastically reduced their workforces. These effects again only emerge for both sectors in Central and Eastern European transition countries and Southern European

countries while, in contrast, Core European countries did not experience any destruction of employment due to jointly introduced process and product innovations.

Finally, *organizational innovations* turn out to be detrimental to employment in the manufacturing sector of Central and Eastern European countries only which suffered nonnegligible employment losses. In contrast, employment appears to be unresponsive to organizational change in both sectors of Southern European EU-member countries and Core European countries. The losses of employment observed in the manufacturing sector of Central and Eastern European transition countries may result from the widespread implementation of labor-saving managerial organizational innovations like improved knowledge management systems intended to ease the use and exchange of information, knowledge and skills or relational organizational innovations intended to modify external relations with other firms or organizations via, for example, outsourcing or subcontracting which tend to destroy jobs. On the other hand, capital-saving organizational innovations appear to have dominated organizational innovations in Southern European countries and Core European countries, leaving employment unaffected in both sectors.

All in all, the results emphasize that different types of innovation trigger similar employment effects in both sectors of Central and Eastern European transition countries and Southern European countries, partly creating, partly destroying employment. In Core European countries, however, employment only reacts to successfully introduced product innovations while process innovations or non-technological organizational innovations entail no net employment effect whatsoever.

7. References

Antonelli, G., R. Antonietti and G. Guidetti, 2010, Organizational Change, Skill Formation, Human Capital Measurement: Evidence from Italian Manufacturing Firms, Journal of Economic Surveys, 24(2), 206-47.

Becker S.O. and P.H. Egger, 2007, Employment Effects of Product and Process Innovations, mimeo.

Benavente, J.M. and R. Lauterbach, 2008, Technological innovation and employment: complements or substitutes?, The European Journal of Development Research, 20(2), 318–29.

Cozzarin, B. and J. Percival, 2006, Complementarities between Organizational Strategies and Innovation, Economics of Innovation and New Technology, 15, 195-217.

Damanpour, F., 1991, Organizational Innovation: A Meta-Analysis of Effects of Determinants and Moderators, Academy of Management Journal, 34(3), 555-90.

Damanpour, F. and S. Gopalakrishnan, 2001, The Dynamics of the Adoption of Product and Process Innovations in Organizations, Journal of Management Studies, 38(1), 45-65.

Evangelista, R., 2000, Sectoral Patterns of Technological Change in Services, Economics of Innovation and New Technologies 9, 183-21.

Evangelista, R. and A. Vezzani, 2010, The employment impact of technological and organizational innovations in European firms, Paper presentato al Workshop "Imprese, innovazione, politiche. Analisi su imprese, settori e performance", Università di Roma Tre, Facoltàdi Economomia, 14 aprile 2010.

Evangelista, R. and M. Savona, 2003, Innovation, Employment and Skills in Services: Firm and Sectoral Evidence, Structural Change and Economic Dynamics, 14(4), 449-74.

Falk, M., 1999, Technological Innovations and the Expected Demand for Skilled Labour at the Firm Level, ZEW Discussion Paper No. 99-59.

Faria, P. and F. Lima, 2009, Firm Decision on Innovation Types: Evidence on Product, Process and Organizational Innovation, Paper presented at the Druid Summer Conference 2009.

Garcia, A., J. Jaumandreu, and C. Rodriguez, 2002, Innovation and Jobs: Evidence from Manufacturing Firms, Discussion Paper, Universidad Carlos III de Madrid.

Greenan, N., 1995, Technologie, changement organisationnel, qualifications et emploi: une etude empirique sur l'industrie manufacturière, INSEE Document de travail G9504.

Greenan, N. and D. Guellec, 2000, Technological Innovation and Employment Reallocation, Labour, 14(4), 547-90.

Hall, B.H., F. Lotti and J. Mairesse, 2008, Employment, innovation, and productivity: evidence from Italian microdata, Industrial and Corporate Change, 17(4), 813-39.

Harrison, R., J. Jaumandreu, J. Mairesse, and B. Peters, 2008, Does Innovation Stimulate Employment? A Firm-Level Analysis Using Comparable Micro-Data from four European Countries, NBER Working Paper 14216.

Jaumandreu, J., 2003, Does Innovation Spur Employment? A Firm-Level Analysis using Spanish CIS Data, Universidad Carlos III de Madrid, mimeo.

Lachenmaier, S. and H. Rottmann, 2006, Employment Effects of Innovation at the Firm Level, IFO Working Papers.

Mantovani, A., 2006, Complementary between Product and Process Innovation in a Monopoly setting, Economics of Innovation and New Technology, 15(3), 219-34.

Marx, Karl, 1977, Capital: Capital. A Critique of Political Economy, Volume I Book One: The Process of Production of Capital, Moscow: Progress Publications. Meriküll, J., 2008, The Impact of Innovation on Employment: Firm- and Industry-level Evidence from Estonia, Eesti Pank Bank of Estonia Working Paper Series 1/2008.

Nickell, S., D. Nicolitsas and M. Patterson, 2001, Does Doing Badly Encourage Management Innovation?, Oxford Bulletin of Economics and Statistics, vol. 63(1), 5-28.

Peters, B., 2004, Employment Effects of Different Innovation Activities: Microeconometric Evidence, ZEW Discussion Paper No. 04-73.

Polder, M., G. van Leeuwen, P. Mohnen and W. Raymond, 2010, Product, Process and Organizational Innovation: Drivers, Complementarity and Productivity Effects, UNU-MERIT Working Paper # 2010-035.

Regev, H., 1998, Innovation, skilled labour, technology and performance in Israeli industrial firms, Economics of Innovation and New Technology, 5, 301-23.

Ross, D.R. and K. Zimmerman, 1993, Evaluating reported determinants of labour demand, Labour Economics, 1, 71-84.

Rottmann, H. and M. Ruschinski, 1998, The Labour Demand and the Innovation Behaviour of Firms, Jahrbücher für Nationalökonomie und Statistik, 217(6), 741-52.

Say, J-B, 1964, A Treatise on Political Economy or the Production, Distribution and Consumption of Wealth, New York: Kelley.

Schmidt, T. and C. Rammer, 2007, Non-technological and Technological Innovation: Strange Bedfellows?, ZEW Discussion Paper No. 07-052.

Sirilli G. and R. Evangelista, 1998, Technological Innovation in Services and Manufacturing: Results from Italian Surveys, Research Policy 27(9), 881-99.

Smolny, W., 1998, Innovations, Prices, and Employment - A Theoretical Model and an Empirical Application for West-German Manufacturing Firms, Journal of Industrial Economics, XLVI(3), 359-81.

Van Reenen, J., 1997, Employment and Technological Innovation: Evidence from U.K. Manufacturing Firms, Journal of Labor Economics, 15(2), 255-84.

8. Appendix

8.1 Regressors

New products – sales growth: Rate of change of a firm's turnover due to newly introduced products.

Process innovations: Dummy which is 1 if the firm introduced either new or significantly improved methods of manufacturing or producing goods and services, new or significantly improved logistics, delivery or distribution methods for inputs, goods or services or new or significantly improved supporting activities for processes, such as maintenance systems or operations for purchasing, accounting, or computing.

Process innovations only: Dummy which is 1 if the firm implemented a process innovation only.

Product and process innovations: Dummy which is 1 if the firm implemented both product and process innovations.

Organizational innovations: Dummy which is 1 if the firm introduced either new or significantly improved knowledge management systems, a major change to the organization of work or new or significant changes in the relations with other firms or public institutions.

Country dummies: Czech Republic, Denmark, Estonia, France, Greece, Hungary, Italy, Latvia, Luxembourg, Portugal, Slovenia, Slovakia, Spain, Sweden.

Industry dummies:

-	
15-37	MANUFACTURING
15-16	Food products; beverages and tobacco
17-19	Textiles and textile products; manufacture of leather and leather products
20-22	Wood and wood products, manufacture of pulp, paper and paper products, publishing and printing
23-24	Coke, refined petroleum products and nuclear fuel; chemicals, chemical products and man-made fibres
25-26	Rubber and plastic; other non-metallic mineral products
27-28	Basic metals; fabricated metal products, except machinery and equipment
29	Machinery and equipment n.e.c
30-33	Office machinery and computers; electrical machinery and apparatus n.e.c; radio, television and communi-
	cation equipment and apparatus; medical, precision and optical instruments, watches and clocks
34-35	Motor vehicles, trailers and semi-trailers; other transport equipment
36-37	Furniture; manufacturing n.e.c; Recycling
51-74	SERVICES
51	Wholesale trade and commission trade, except of motor vehicles and motorcycles
60-63	Transport; Supporting and auxiliary transport activities
64	Post and telecommunications
65-67	Banking and insurance
72	Computer and related activities
73	Research and Development
74.2	Architectural and engineering activities and related technical consultancy
74.3	Technical testing and analysis

8.2 Instruments

New to market: Market share of goods and services introduced during 2002 and 2004 that were new to the firm's market.

Patent: Dummy which is 1 if the firm applied for a patent between 2002 and 2004.

Science: Dummy which is 1 if institutional sources like universities or other higher education institutions or government or public research institutes were of 'high' or 'medium' importance as sources of information for a firm's innovation activities.

R&D intensity: Ratio of total (intramural and extramural) R&D expenditures to firm turnover in 2004.

Continuous innovator: Dummy which is 1 if, during 2002 and 2004, the firm performed R&D continuously.

Table 3

Effects of product and process innovations on employment in Manufacturing and Services, by country-groups (GMM)

Manufacturing			Services					
Country group	ALL	CEEC	CORE	SOUTH	ALL	CEEC	CORE	SOUTH
Method	GMM	GMM	GMM	GMM	GMM	GMM	GMM	GMM
Dependent variable		l – (g	$_1 - \widetilde{\pi}_1)$		$I-(g_1-\widetilde{\pi}_1)$			
Sales growth - new products	0.632	0.655	0.409	0.664	0.695	0.903	0.904	0.431
	(8.09)***	(6.41)***	(2.16)**	(5.41)***	(7.23)***	(4.43)***	(4.70)***	(3.42)***
Process innovations only	2.355	2.801	-0.540	3.803	1.664	3.156	-0.900	3.436
	(3.61)***	(1.57)	(0.49)	(4.51)***	(1.78)*	(1.25)	(0.69)	(2.25)**
Process and product innovations	-5.222	-7.657	-1.347	-6.590	-5.919	-9.388	-1.495	-10.182
	(8.63)***	(5.70)***	(1.58)	(7.06)***	(5.79)***	(3.84)***	(1.03)	(5.69)***
Organizational innovations	-1.199	-2.066	-0.837	-0.873	0.506	0.521	-0.480	1.842
	(2.58)***	(1.99)**	(1.15)	(1.29)	(0.72)	(0.34)	(0.45)	(1.54)
No. of observations	25,809	7,502	7,006	11,301	17,083	3,798	7,336	5,949
Hansen J-test p-value	5.219	3.985	4.107	5.272	0.137	1.878	8.332	4.300
degrees of freedom	(3)	(3)	(3)	(3)	(2)	(2)	(2)	(2)
p-value	0.1565	0.2631	0.2502	0.1530	0.9339	0.3910	0.0155	0.1165

Notes: All specifications for the manufacturing sector use new to the market, R&D intensity, patent, science as instruments while all specifications for the service sector use new to the market, patent and continuous innovator as instruments. Country and industry dummies are included in all specifications.

The pooled sample (ALL) consists of the Czech Republic, Denmark, Estonia, France, Greece, Hungary, Italy, Latvia, Luxembourg, Portugal, Slovenia, Slovakia, Spain and Sweden.

Central and Eastern European transition countries (CEEC) comprise the Czech Republic, Estonia, Hungary, Latvia, Slovenia and Slovakia.

The group of Core European countries (CORE) comprises Denmark, France, Luxembourg and Sweden.

Southern European countries (SOUTH) comprise Greece, Italy, Portugal and Spain.

Country and industry dummies are included in all regressions.

Absolute value of t statistics in parentheses

* significant at 10%; ** significant at 5%; *** significant at 1%

Short list of the most recent wiiw publications (as of January 2011

For current updates and summaries see also wiiw's website at <u>www.wiiw.ac.at</u>

Change Begets Change: Employment Effects of Technological and Non Technological Innovations – a Comparison across Countries

by Sandra M. Leitner, Johannes Pöschl and Robert Stehrer

wiiw Working Papers, No. 72, January 2011 23 pages including 3 Tables hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

wiiw Monthly Report 12/10

edited by Leon Podkaminer

- A structural decomposition of international trade
- · Migration, skills and productivity
- Structural root causes of the debt crises/heterogeneity
- Statistical Annex: Selected monthly data on the economic situation in Central and Eastern Europe

wiiw, December 201022 pages including 10 Tables and 3 Figures(exclusively for subscribers to the wiiw Service Package)

The role of services in the New Member States: A comparative analysis based on inputoutput tables

by Doris Hanzl-Weiss and Robert Stehrer

wiiw Research Reports, No. 366,November 2010 91 pages including 26 Tables and 37 Figures hardcopy: EUR 24.00 (PDF: EUR 15.00)

wiiw Monthly Report 11/10

edited by Leon Podkaminer

- Albania: agnostic growth forecast
- · Bosnia and Herzegovina: economic recovery and new political constellation
- · Croatia: recovery not yet in sight
- · Kazakhstan: rebound is faster
- · Macedonia: slow and stable
- · Montenegro: recovery delayed
- · Russian Federation: oil-fuelled recovery loses steam
- · Serbia: inflationary pressures
- Turkey: swift return to pre-crisis levels
- Ukraine: the return of high inflation
- Statistical Annex: Selected monthly data on the economic situation in Southeast Europe, Russia and Ukraine

wiiw, November 2010

38 pages including 20 Tables

(exclusively for subscribers to the wiiw Service Package)

Migration, Skills and Productivity

by Peter Huber, Michael Landesmann, Catherine Robinson and Robert Stehrer with Robert Hierländer, Anna Iara, Mary O'Mahony, Klaus Nowotny and Fei Peng

> wiiw Research Reports, No. 365, November 2010 172 pages including 81 Tables and 20 Figures hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

Inequality, Growth and Public Spending in Central, East and Southeast Europe

by Mario Holzner

wiiw Working Papers, No. 71, October 2010 15 pages including 3 Tables hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

wiiw Monthly Report 10/10

edited by Leon Podkaminer

- · Bulgaria: muddling through
- · Czech Republic: changing structure of growth
- · Hungary: slow recovery based on external demand
- · Poland: waiting for the investment take-off
- · Romania: recovery postponed
- · Slovakia: robust exports prevailing over weak domestic demand
- · Slovenia: catching up slowly
- · Baltics: : exports and restocking drive growth revival
- Statistical Annex: Selected monthly data on the economic situation in Central and Eastern Europe

wiiw, Octoberr 201042 pages including 17 Tables(exclusively for subscribers to the wiiw Service Package)

The Impact of Preferential Trade Agreements on the Margins of International Trade

by Neil Foster, Johannes Pöschl and Robert Stehrer

wiiw Working Papers, No. 70, September 2010 25 pages including 9 Tables hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

wiiw Monthly Report 8-9/10

edited by Leon Podkaminer

- The intensity of competition in Central, Eastern and Southeast Europe: stylized facts and repercussions for sectoral price developments
- Exchange market pressure contagion in CESEE
- · Scenarios for Ukraine's medium- and long-term development
- · Markets and morals
- Statistical Annex: Selected monthly data on the economic situation in Central and Eastern Europe

wiiw, August-September 2010

36 pages including 10 Tables and 7 Figures

(exclusively for subscribers to the wiiw Service Package)

Discrepancies between Purchasing Power Parities and Exchange Rates under the Law of One Price: A Puzzle (partly) Explained?

by Leon Podkaminer

wiiw Working Papers, No. 69, September 2010 18 pages including 5 Tables hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

An Empirical Characterization of Redistribution Shocks and Output Dynamics

by Klemens Hauzenberger and Robert Stehrer

wiiw Working Papers, No. 68, August 2010 47 pages including 6 Tables and 4 Figures hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

Ukraine, the European Union and the International Community: Current Challenges and the Agenda for Overcoming the Stalemate

by Vasily Astrov, Igor Burakovsky, Grzegorz Gromadzki, Peter Havlik and Vasyl Yurchyshyn

wiiw Research Reports, No. 364, July 2010 93 pages including 27 Tables and 15 Figures hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

Austrian Exporters: A Firm-Level Analysis

by Roman Stöllinger, Robert Stehrer and Johannes Pöschl

wiiw Working Papers, No. 67, July 2010 32 pages including 14 Tables and 4 Figures hardcopy: EUR 8.00 (PDF: free download from wiiw's website)

Will Exports Prevail over Austerity?

by Vasily Astrov, Mario Holzner, Kazimierz Laski, Leon Podkaminer et al.

wiiw Current Analyses and Forecasts. Economic Prospects for Central, East and Southeast Europe, No. 6, July 2010 164 pages including 30 Tables and 33 Figures hardcopy: EUR 80.00 (PDF: EUR 65.00)

wiiw Monthly Report 7/10

edited by Leon Podkaminer

- A few reflections on fiscal and monetary policy in the euro area
- · Unit labour costs, exchange rates and responses to the crisis
- Poland's presidential election: and the winner is ...
- Statistical Annex: Selected monthly data on the economic situation in Southeast Europe, Russia and Ukraine

wiiw, July 2010

28 pages including 9 Tables and 6 Figures

(exclusively for subscribers to the wiiw Service Package)

wiiw Service Package

The Vienna Institute offers to firms and institutions interested in unbiased and up-to-date information on Central, East and Southeast European markets a package of exclusive services and preferential access to its publications and research findings, on the basis of a subscription at an annual fee of EUR 2,000.

This subscription fee entitles to the following package of **Special Services**:

- A free invitation to the Vienna Institute's Spring Seminar, a whole-day event at the end of March, devoted to compelling topics in the economic transformation of the Central and East European region (for subscribers to the wiiw Service Package only).
- Copies of, or online access to, *The Vienna Institute Monthly Report*, a periodical consisting of timely articles summarizing and interpreting the latest economic developments in Central and Eastern Europe and the former Soviet Union. The statistical annex to each *Monthly Report* contains, alternately, country-specific tables or graphs with monthly key economic indicators, economic forecasts, the latest data from the wiiw Industrial Database and excerpts from the wiiw FDI Database. This periodical is not for sale, it can only be obtained in the framework of the wiiw Service Package.
- Free copies of the Institute's **Research Reports** (including **Reprints**), **Current Analyses and Forecasts**, **Country Profiles** and **Statistical Reports**.
- A free copy of the *wiiw Handbook of Statistics* (published in October/November each year and containing more than 400 tables and graphs on the economies of Albania, Bosnia and Herzegovina, Bulgaria, Croatia, the Czech Republic, Estonia, Hungary, Latvia, Lithuania, Macedonia, Montenegro, Poland, Romania, Russia, Serbia, the Slovak Republic, Slovenia and Ukraine)
- Free online access to the wiiw Monthly Database, containing more than 1200 leading indicators monitoring the latest key economic developments in ten Central and East European countries.
- Consulting. The Vienna Institute is pleased to advise subscribers on questions concerning the East European economies or East-West economic relations if the required background research has already been undertaken by the Institute. We regret we have to charge extra for *ad hoc* research.
- Free access to the Institute's specialized economics library and documentation facilities.

Subscribers who wish to purchase wiiw data sets **on CD-ROM** or special publications not included in the wiiw Service Package are granted considerable **price reductions**.

For detailed information about the wiiw Service Package please visit wiiw's website at www.wiiw.ac.at

To The Vienna Institute for International Economic Studies Rahlgasse 3 A-1060 Vienna

- O Please forward more detailed information about the Vienna Institute's Service Package
- O Please forward a complete list of the Vienna Institute's publications to the following address

Please enter me for

- 1 yearly subscription of *Research Reports* (including *Reprints*) at a price of EUR 120.00 (hardcopy, Austria),
 EUR 135.00 (hardcopy, Europe), EUR 155.00 (hardcopy, overseas) and EUR 48.00 (PDF download with password) respectively
- 1 yearly subscription of *Current Analyses and Forecasts* a price of EUR 150.00 (hardcopy, Austria),
 EUR 155.00 (hardcopy, Europe), EUR 170.00 (hardcopy, overseas) and EUR 120.00 (PDF download with password) respectively

Please forward

0	the following issue of Research Reports
0	the following issue of Current Analyses and Forecasts
0	the following issue of Working Papers
0	the following issue of Research Papers in German language
0	the following issue of wiiw Database on Foreign Direct Investment
0	the following issue of wiiw Handbook of Statistics
0	(other)

Name		
Address		
Telephone	Fax	E-mail
Date		Signature

Herausgeber, Verleger	, Eigentümer und Hersteller:
	Verein "Wiener Institut für Internationale Wirtschaftsvergleiche" (wiiw), Wien 6, Rahlgasse 3
	ZVR-Zahl: 329995655
Postanschrift:	A-1060 Wien, Rahlgasse 3, Tel: [+431] 533 66 10, Telefax: [+431] 533 66 10 50
Internet Homepage:	www.wiiw.ac.at
Nachdruck nur auszug	sweise und mit genauer Quellenangabe gestattet.
P.b.b. Verlagspostamt	1060 Wien